Author’s response to reviews

Title: Exploring barriers to seeking health care among Kenyan Somali women with female genital mutilation: a qualitative study

Authors:

Samuel Kimani (tkimani@uonbi.ac.ke)
Caroline Kabiru (carolinekabiru@gmail.com)
Jacinta Muteshi (jmuteshi@popcouncil.org)
Jaldesa Guyo (jaldesagw@yahoo.com)

Version: 1 Date: 31 May 2019

Author’s response to reviews:

The comment has been addressed in the manuscript in the method section on page 8 line 164-168 and on ethical consideration section on page 22 & 23 lines 490-493. The modified statement reads as below:

Study participants aged 18 years and older granted informed consent, while assent was obtained from younger (minors) participants, in which explanation on the study was done but the consent was provided by the parents or legal guardian for participation into the study and for audio recording of the discussion/interviews.